Author's response to reviews

Title: Association of IL1A and IL1B loci with primary open angle glaucoma

Authors:

Suddhasil Mookherjee (suddhasilmookherjee@gmail.com)
Deblina Banerjee (banerjee.deblina@gmail.com)
Subhadip Chakraborty (subhadipiicb@gmail.com)
Antara Banerjee (banerjee.antara77@gmail.com)
Indranil Mukhopadhyay (indranilm100@gmail.com)
Abhijit Sen (skeya063@gmail.com)
Kunal Ray (kunalray@gmail.com)

Version: 7 Date: 16 June 2010

Author’s response to reviews: see over
Dear Dr. Edmunds,

We have completed the proofreading of the manuscript entitled “Association of IL1A and IL1B loci with primary open angle glaucoma”. All suggested major and the minor changes have been made.

**MAJOR REVISIONS:**

**References:** The referencing has been formatted according to the journal rule.

**MINOR REVISIONS:**

Authors’ qualifications or job titles have been removed from the manuscript and submissions system.

We hope that the editors will find our manuscript suitable for publication in BMC Medical Genetics.

Sincerely,
Kunal Ray